**Author’s response to reviews**

**Title:** The Societal Economic Burden of Autosomal Dominant Polycystic Kidney Disease in the United States

**Authors:**

Martin Cloutier (Martin.Cloutier@analysisgroup.com)
Ameur Manceur (Ameur.Manceur@analysisgroup.com)
Annie Guerin (Annie.Guerin@analysisgroup.com)
Myrlene Sanon Aigbogun (Myrlene.Sanon@otsuka-us.com)
Dorothee Oberdhan (Dorothee.Oberdhan@otsuka-us.com)
Marjolaine Gauthier-Loiselle (marjolaine.gauthier-loiselle@analysisgroup.com)

**Version:** 1  
**Date:** 27 Jun 2019

**Author’s response to reviews:**

1. **Overlap**  
Response: Sections with any resemblance to our previous publication have been revised and reworded.

2. **Funding**  
Response: Role of funding body has been described

3. **Authors' Contributions**  
Response: Individual contributions have been added

4. **Formatting**  
Response: figure legends have been added under 'Figure legends' after references